Author's response to reviews

Title: When children play, they feel better: Organized activity participation and health in adolescents

Authors:

Petr Badura (petr.badura@upol.cz)
Andrea M Geckova (andrea.geckova@upjs.sk)
Dagmar Sigmundova (dagmar.sigmundova@upol.cz)
Jitse P van Dijk (j.p.van.dijk@umcg.nl)
Sijmen A Reijneveld (s.a.reijneveld@umcg.nl)

Version: 3 Date: 11 June 2015

Author's response to reviews: see over
June 11, 2015

Diana Marshall
Senior Managing Editor
BMC Public Health
263 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Dear Dr. Marshall,

upon the e-mail from Victorino Silvestre of the 9th June, please find enclosed the second revision of our manuscript entitled: "When children play, they feel better: Organized activity participation and health in adolescents" (MS: 1280922056174184) by Petr Badura, Andrea Madarasova Geckova, Dagmar Sigmundova, Jitse P. van Dijk and Sijmen A. Reijneveld, which we are submitting for exclusive consideration of publication in BMC Public Health. Below please find the response to required revision:

1) Please state in the manuscript whether all the data used for your study is openly available or whether you obtained permission (and from who) to use this.

Response: The data for the submitted study are not openly available, therefore the two following sentences were added to the Methods, subsection Procedure - page 5, line 25:
“The data used in this study is not publicly available. Permission to use it was obtained from the Czech HBSC Principal Investigator.”

We would like to thank you for considering this manuscript for publication in BMC Public Health and hope it will be of interest.

Yours sincerely,
on behalf of all co-authors,

Petr Badura, MSc.

petr.badura@upol.cz; petr.badura@hbsc.org
office: +420 585 636 072; cell phone: +420 777 703 154